Author's response to reviews

Title: Small primary adenocarcinoma in adenomyosis with nodal metastasis: a case report

Authors:

Giacomo Puppa (gpuppa@yahoo.com)
Makio Shozu (shozu@med.kanazawa-u.ac.jp)
Tiziana Perin (tperin@cro.it)
Kazuhito Nomura (nomura@med.kanazawa-u.ac.jp)
Annunziata Gloghini (agloghini@cro.it)
Elio Campagnutta (ecampagnutta@cro.it)
Vincenzo Canzonieri (vcanzonieri@cro.it)

Version: 5  Date: 16 April 2007

Author's response to reviews: see over
MS: 6282855291225588
Small primary adenocarcinoma in adenomyosis with nodal metastasis: a case report

To the Editor BMC Cancer
We are proud that our ms is accepted in BMC Cancer for publication.

- Change affiliation and address for one author.
  Professor Shozu: Graduate School of Chiba University, Reproductive Medicine, Chuo-ku, Chiba, Japan. shozu@faculty.chiba-u.jp

- Written consent: already sent
  On 20/11/2006, after I originally submitted the ms, I received the following request:

Dear Dr Puppa

BioMed Central welcomes useful and well-described reports of exceptional or unusual cases. To help us better focus attention on the specific merits of your manuscript, could you please answer the following questions.

* Please describe how this case report is exceptional or unusual
* Please describe the educational and/or scientific merit of the report
* Does the case report include an up-to-date list and review of all previously published cases in the field?
* Has written consent been obtained from the patient or their relative for publication of this report? We will need a copy posted or faxed to this office before we can consider publication of your manuscript.

As we are unable to proceed with peer review until such time as we have an answer to these questions, I would urge you to respond as soon as possible.

Best wishes
Chantal
Chantal Botha
Senior Editorial Administrator, BMC-series Journals
Editorial Department
BioMed Central Ltd
Middlesex House

I therefore sent via FAX a letter answering to the questions one by one, together with a copy of the written consent. After I sent such documents, the reviewing process started. However, if necessary, I can send again a copy of the written consent.

- Query
  Abstract- Please change the last subheading to Conclusions.
  done

  Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT.
  done
Acknowledgements- Please acknowledge receipt of consent form the patient for the publication of the case.
   done

References- Please place the references after the Acknowledgements.
   done

Thank you      Giacomo Puppa.